Author’s response to reviews

Title: The Effects of apoptosis vulnerability markers on the myocardium in depression after myocardial infarction

Authors:

Yiming Wang (yimingw66@yahoo.com)
Xingde Liu (lxd@gmc.edu.cn)
Dongfeng Zhang (53588@163.com)
Jianhui Chen (416494836@qq.com)
Shuzheng Liu (1057386401@qq.com)
Michael Berk (mikebe@barwonhealth.org.au)

Version: 7 Date: 12 December 2012

Author’s response to reviews: see over
Dear Sabina

Many thanks for your assistance. The following responses to editor’s comments have now been addressed.

**Answer questions:**

1. Please ensure that each author affiliation contains, in the following order, department, institution, street name and number, town, post code and country.

   Each author affiliation contains has been added in the first page, highlighted in red.

2. Please ensure that Table 2 is split into multiple columns using the tables tool. It is currently a single column table with the text separated by spaces/tabs. This will ensure that the table content remain aligned during any further editing.

   We have reedited the Table 2 using the tables tool.

3. Please ensure that reference 25 is cited (in numerical order).

   Yes, reference 25 has been cited in methods, page 7, lines 11 and page 9, lines 3.

4. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

   Typographical errors have been checked and corrected.

12 Dec 2012
Michael Berk cont’d…

Yours sincerely

Michael Berk
Professor of Psychiatry, Deakin University